Author’s response to reviews

Title: Surgical management of pyelo-ureteral junction syndrome in a resource-limited setting: Case of Zinder National Hospital, Niger

Authors:
Harissou Adamou (harissou1976@yahoo.fr)
Maazou Halidou (halidou3@yahoo.fr)
Ibrahim Amadou Magagi (imagagi@gmail.com)
Hassane Diongolé (diongolen@yahoo.fr)
Mahamadou Doutchi (m.doutchi@yahoo.fr)
Oumarou Habou (bhomar70@yahoo.fr)
Kabirou Ganiou (adniyinakeji@yahoo.fr)
Amadou Soumana (drsoumanaamadou@gmail.com)

Version: 1 Date: 07 Aug 2018

Author’s response to reviews:

Dear editor,

We have taken into account your remarks and have brought to the manuscript all the answers requested. Thank you for your interest in our work.